Author's response to reviews

**Title:** Acute abdomen and renal failure due to bilateral renal infarction in a woman: a case report

**Authors:**

khaireddine bouassida (khayri14@hotmail.com)
wissem hmida (khayri14@hotmail.com)
amira zairi (khayri14@hotmail.com)
adnen hidoussi (khayri14@hotmail.com)
faouzi mallat (khayri14@hotmail.com)
mehdi jaidane (khayri14@hotmail.com)
khaled hani (khayri14@hotmail.com)
faouzi ali mosbah (khayri14@hotmail.com)

**Version:** 2  **Date:** 14 September 2011

Author's response to reviews:

Sousse September 14, 2011

From: Khaireddine Bouassida
Department of urology
Hospital of Sahloul, Sousse Tunisia

To Journal of Medical Case Report

Subject: Letter to the editor

Thank you very much for all these comments you’ve made in reviewing our study. Taking in to account remarks provided by you, we modified our article to make it appropriate for publication, hopefully.

1- A patient consent section was included to the article submission.
2- An author’s contributions section was written.
3- A competing interest section was added.
4- The title of the manuscript was changed according to the structure of the journal.
5- The ethnicity of the patient was included as well in the abstract as in the case presentation section of the manuscript.

Best regards

Khaireddine Bouassida